Author's response to reviews

Title: Successful resolution of symmetrical peripheral gangrene after severe acute pancreatitis: a case report

Authors:

Chen-Yi Liao Dr. (ericliao0217@gmail.com)
Shih-Chung Huang Dr. (sghung@hotmail.com)
Cheng-Hui Lin Dr. (beetle95@gmail.com)
Ren-Jy Ben Dr. (benrbridge@gmail.com)
Ching-Chang Le Dr. (tsgh11@gmail.com)
Mei-Yu Liu Miss (lancer0106@yahoo.com.tw)

Version: 6
Date: 31 July 2015

Author's response to reviews: see over
Dear Editor-in-Chief,

Please find enclosed the manuscript: “Successful resolution of symmetrical peripheral gangrene after severe acute pancreatitis: a case report”, by Chen-Yi., et al., to be submitted as a case report to the Journal of medical case reports. The case should raise the interest in the field of gastroenterology due to following reasons. We had made corrections according to the reviewers response. We also seek for English specialist for better English grammar modification. We had made response to the reviewers and the editors.

We look forward to hearing from you at your earliest convenience.

Sincerely yours,

Chen-Yi., et al. on behalf of the authors.

Corresponding author: Chen-Yi liao at Kaohsiung Armed Forces General Hospital Zhongzheng 1st Rd., Lingya Dist., Kaohsiung City, Taiwan (R.O.C.)

ericliao0217@gmail.com, phone number: +886-07-7490633 Personal Phone number: 0929674696